

Response to Reviewer

Unrealistic target journals, poor paper construction, poor research design, lack of novelty, irrelevant clinical research, unrealistic time-frames, improper English, plagiarism--these are some of the many aspects of writing that you must steer clear if you want your manuscript to be selected.

At Pubrica, experienced editors show you the way forward and guide you in this cumbersome process; experts advise you from conceptualizing novel ideas to responding to your reviewers. We are with you till you publish your journal successfully.

How our experts address the reviewer comments?

> Carefully gathering thoughts: Browse the comments and carefully gather your thoughts before addressing them. Then

take a fresh look at the comments and determine what modifications the reviewers'/editors' want to see in your manuscript.

- > Track changes: Use track changes or different colour to clarify revised text in the manuscript
- Rebuttal letter: (1) Attach a separate sheet listing response to referee comments (2) Include referee comments in response (3) Respond to individual comments (4) State upfront what action was taken (done /not done) and then explain the reason
- Address each reviewer's comments separately but on one response note as reviewer A and B's response to any comment. *the reviewercomment does not include formatting, and resubmission Help

Pubrica Offerings

Pubrica offers you complete publishing support across a variety of publications, journals, and books. You can now morph your concepts into incisive reports with our array of writing services: regulatory writing, Clinical Report Forms (CRF), biostatistics, manuscripts, business writing, physician reports, medical writing and more. Experts in Science, Technology, Engineering and Mathematics (STEM), and pundits with therapeutic repertoire. Publishing that medical paper or getting a regulatory drug approval is now easy. Save time and money through Pubrica's support.

Related Services

- Publication Support
- Physician Writing
- Scientific Communication

We serve for



Pharmaceutical Companies



Medical Devices



Authors A

-For Plans and Packages click here-

ABOUT US

Pubrica Scientific Writing & Publication offers comprehensive medical, scientific, technical, and business writing services. We offer an array of writing services: manuscript writing, regulatory writing, Clinical Report Form (CRF) writing, biostatis tical report writing, academic and business writing, physician writing, medical writing and more. Subject-Matter Experts (SMEs) in Science, Technology, Engineering, and Mathematics (STEM), and pundits with therapeutic repertoire. Publishing that medical paper or getting a regulatory drug approval is now easy. We assist you in every phase of your project.